
diff -r c90f803dc7ea -r eb2004430215 handouts/ho05.tex
--- a/handouts/ho05.tex	Sat Dec 20 14:48:15 2014 +0000
+++ b/handouts/ho05.tex	Sat Dec 27 04:10:36 2014 +0000
@@ -811,6 +811,16 @@
 
 \subsubsection*{Further Reading}
 
+A blogpost that describes the first few milliseconds of an HTTPS connection
+is here
+
+\begin{center}
+\url{http://www.moserware.com/2009/06/first-few-milliseconds-of-https.html}
+\end{center}
+
+\noindent
+it disentangles every message sent between a client and a server.
+
 If you want to know more about how cars can be hijacked,
 the paper  
 




